Author's response to reviews

Title: Type and etiology of liver cirrhosis are not related to the presence of hepatic encephalopathy or health-related quality of life: a cross-sectional study

Authors:

Evangelos Kalaitzakis (evangelos.kalaitzakis@vgregion.se)
Axel Josefsson (gusjoaxe@student.gu.se)
Einar Björnsson (einar.bjornsson@medic.gu.se)

Version: 5 Date: 8 October 2008

Author's response to reviews:

Dear editor,

One small change has been made in the manuscript. In page 6 (methods section) the end of the sentence "Although a history of alcohol abuse is lacking in these patients it is difficult to exclude with certainty an alcoholic etiology in some of these patients" was altered. This sentence now appears as "Although a history of alcohol abuse is lacking in these patients it is difficult to exclude with certainty an alcoholic etiology in some."

Yours truly,

Evangelos Kalaitzakis MD, PhD